Author's response to reviews

Title: Intravenous leiomyomatosis of the uterus with extension to the right heart

Authors:

Yu-Feng Lou (longlou2000@163.com)
Xin-Ping Shi (Eric_hangzhou@163.com)
Ze-Zhou Song (zezhou_song@126.com)

Version: 2 Date: 15 September 2011

Author's response to reviews:

Dear Editor:

Thank you very much for your kind letter along with your and the reviewers’ comments. I greatly appreciate you and the reviewers for your support of my manuscript entitled “Intravenous leiomyomatosis of the uterus with extension to the right heart” to be published in “Cardiovascular ultrasound”. According to the suggestions and comments given by you and the reviewers I have revised the manuscript. My point-by-point response to your and the reviewers’ comments attached below. I would appreciate you very much if you let us know your decision at your earliest convenience. Wish this work could appear in “Cardiovascular ultrasound” soon. Thank you!

REVIEWER Comments:

Comments 1. Please highlight the fact, in medical history, that the patient underwent hysterectomy.

Answer: I have highlighted the fact that the patient underwent hysterectomy at the section of case report and discussion.

Comments 2. Please try to describe the mechanisms of diffusion to the heart. Incomplete surgery?

Answer: I have described the mechanisms of diffusion to the heart at the section of discussion.

Comments 3. There are many similar cases already published, some of them coming from China. Please in revising the literature try to cite and cover most of the publications on the same topic.

Answer: I have cited and covered most of the publication on the same topic at the section of discussion.

Comments 4. In line with the previous comment it would be interesting to have a table of the published cases in China.

Answer: I have had a table of the published cases in China.

Comments 5. The figures are not particularly self-explanatory and well defined. It would be important to have clearer images, even clips since there is no space
limit.
Answer: I absolute agreed with the reviewer. However, the clips were not usually stored, therefore, I am very sorry that the case is without the clips. In addition, the images which we submitted were best to relate the case. I am very sorry to say that.

Comments 6. Please comply to journal rules for references.
Answer: I have complied to journal rules for references.